Author's response to reviews

Title: Identification of a deep intronic mutation in the COL6A2 gene by a novel custom oligonucleotide CGH array designed to explore allelic and genetic heterogeneity in collagen VI-related myopathies

Authors:

Matteo Bovolenta (bvlmtt@unife.it)
Marcella Neri (nremcl@unife.it)
Elena Martoni (elena.martoni@unife.it)
Anna Urciuolo (anna.urciuolo@unipd.it)
Patrizia Sabatelli (psabatel@area.bo.cnr.it)
Marina Fabris (marina.fabris@unife.it)
Paolo Grumati (paolo.grumati@unipd.it)
Eugenio Mercuri (mercuri@rm.unicatt.it)
Enrico Bertini (ebertini@tin.it)
Luciano Merlino (luciano.merlino@unife.it)
Paolo Bonaldo (bonaldo@bio.unipd.it)
Alessandra Ferlini (fla@unife.it)
Francesca Gualandi (gdf@unife.it)

Version: 10 Date: 15 February 2010

Author's response to reviews:

Dear Editor please find enclosed the manuscript entitled

“Identification of a deep intronic mutation in the COL6A2 gene by a novel custom oligonucleotide CGH array designed to explore allelic and genetic heterogeneity in Collagen VI-related myopathies.”

We are submitting the manuscript again because we are experiencing a problem during the final formatting stage. We kindly asked to Assistant Scientific Editor Danielle Burgess if it is possible to accept the paper as it is.

Thanking you in advance for your attention

Sincerely Yours

Francesca Gualandi

Francesca Gualandi, MD, PhD
Dipartimento di Medicina Sperimentale e Diagnostica
Sezione di Genetica Medica
Via Fossato di Mortara, 74
44100 Ferrara
Italy
Tel: +39-532-974497
Fax: +39-532-236157
e-mail: gdf@unife.it